Author's response to reviews

Title: Outcome measurement in clinical trials for Ulcerative Colitis: Toward standardisation

Authors:

Rachel M Cooney (rachel.cooney@clinical-medicine.oxford.ac.uk)
Bryan F Warren (bryan.warren@orh.nhs.uk)
Douglas G Altman (doug.altman@cancer.org.uk)
Maria T Abreu (maria.abreu@mssm.edu)
Simon PT Travis (simon.travis@ndm.ox.ac.uk)

Version: 5 Date: 9 June 2007

Author's response to reviews: see over
Cover letter re changes made 8.6.07

Title changed to sentence case
Author list- initial M added to Rachel M Cooney
Simon PT Travis corrected to Simon PL Travis
Abstract title added
Abstract added. Final sentence from initial submission removed.
All table placement texts removed from body of manuscript ie Table 1-5
References- et al removed from all references and authors included in full
Conflict of interest changed to competing interests.